
Allen, Pam, NMENV 

From: 
Sent: 
To: 
Subject: 

Maestas, Ricardo, NMENV 
Wednesday, August 13, 2014 11:10 AM 
Allen, Pam, NMENV 
FW: WIPP Information - For Call Today 

Attachments: North Rib Waste Face.docx; 41-b-857 WHT dp Daily Averages_4-28-14.pdf; 41-b-856 
WHT dp Daily Averages_4-28-14.pdf; Station A and B Filter Readings for Public Release 
4-28-14.xlsx; Environmental Sampling 2014-04-28 1430_dg_bw.xlsx 

April 

From: Kliphuis, Trais, NMENV 
Sent: Tuesday, April 29, 2014 1:15 PM 
To: Flynn, Ryan, NMENV; Kendall, Jeff, NMENV 
Cc: Winchester, Jim, NMENV; Tongate, Butch, NMENV; Schwender, Erika, NMENV; Blaine, Tom, NMENV; Skibitski, 
Thomas, NMENV; Kieling, John, NMENV; Holmes, Steve, NMENV; Maestas, Ricardo, NMENV; Smith, Coleman, NMENV; 
LucasKamat, Susan, NMENV; Turner, Jill, NMENV; Nelson, Morgan, NMENV; Ines Triay (triayin@fiu.edu) 
Subject: FW: WIPP Information - For Call Today 

From: Oba Vincent [mailto:oba.vincent@cbfo.doe.gov] 
Sent: Tuesday, April 29, 2014 10:52 AM 
To: 'Pace, Berry'; Kliphuis, Trais, NMENV; 'peake.tom@epa.gov'; 'Edwards, Jonathan'; 'Walsh, Jonathan'; 'Perrin, Alan'; 
'Kehrman, Bob - RES'; 'Chavez, Rick - RES'; 'Stone.Nick@epa.gov'; Smith, Coleman, NMENV; 
'brozowski.george@epa.gov'; 'Fraass, Ron'; 'Hardy, Russell'; 'Veal.Lee@epamail.epa.gov'; 'Economy, Kathleen 
(Economy.Kathleen@epa.gov)'; 'Poppell, Sam W. (Poppell.Sam@epa.gov)'; Maestas, Ricardo, NMENV 
Cc: George Basabilvazo - WIPPNet; 'Reynolds, Tammy - NWP'; 'Harris, Alton - DOE EM'; Susan McCauslin; 'Joe Harvill 
(jharvill@portageinc.com)'; 'Kennedy, Scott - NWP'; 'Jones, Stewart - RES'; 'Oates, Berta - CTAC'; 
'schultheisz.daniel@epa.gov'; Philip Theisen - ORISE; Russ Patterson - WIPPNet; 'Kouba, Steve - WRES'; Roger Nelson -
WIPPNet; 'Bignell, Dale - CTAC'; Susan McCauslin - WIPPNet; Anthony Stone - WIPPNet; J.R. Stroble - WIPPNet 
Subject: WIPP Information - For Call Today 

Attached are the filter performance data, environmental sample results, action items listing and Station A/B data. Also 
included is a picture of the waste taken in Panel 7. 

Below is a summary of activities: (Please note, activities and dates are subject to change. Please verify the most 
current dates of any information provided). 

• Full video is 10 GigaBites - What was posted is "meat of it" 

• 34 additional air sampling results from Apr 15 and Apr 22 were reported. This included the first air sample 

results from Artesia, Eunice, Hobbs and Loving. Six soil samples were received from the 29 taken which 

transverse the original plume (13 of 29 results reported). All results were below the MDC. The sample collected 

from Mills Ranch on 2/20/14 was recounted. This was the only soil sample which was above the MDC. The new 

result is slightly higher but still consistent with results seen for Pu-239 in this area prior to the release. 

• Underground remains in Filtration Mode utilizing Fan 860B. Plans on Thursday are to switch to Fan 860C and 

conduct PMs on Fans B&C 

• Cleaning of the soot from the fire event in the Waste Handling Tower is continuing; Cleaning of the 5th floor is 

complete. Cleaning of the 4th floor continues. The 3rd floor is being initiated. 

• Weekly MSHA inspections of the AIS & SS are being performed today. 
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• U/G Ventilation filter change is scheduled for the week of May 12. This includes building a containment area. 

• The next mine entry is planned for Wednesday, 4/30. During this entry, additional photos will be taken and 

filter paper exchanged from samplers in Panel 7. An additional entry is planned for the end of the week {Fri or 

Sat) and may involve the collection of samples. {No Change) Plan is to have phone line open. Will use same 

phone line as before. 

• 10 Shipments to WCS are scheduled for the week. The weekly summary of waste being prepared for shipment 

was posted on the Portal. {No Change) 

• 15 day RCRA Contingency Plan Report was submitted on 4/28. 

• Entry will start about 9:00-9:30 AM tomorrow. 

• Still promising to send procedure for tomorrow's re-entry by this afternoon. 

NOTE CALL-IN NUMBER: 1888 413 3490, Code 7175394 
Thank you 

Oba 
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